Author’s response to reviews

Title: Evaluation of pharmacotherapy complexity in residents of long-term care facilities: a cross-sectional descriptive study

Authors:
Vanessa Alves-Conceição (vanessa1_alves@hotmail.com)
Daniel Silva (danieltenorio.lepfs@hotmail.com)
Vanessa Santana (vanessa-farmaceutica@hotmail.com)
Edileide Santos (edileidegsantos@hotmail.com)
Lincoln Santos (marques.lincoln@homail.com)
Divaldo Lyra Júnior (lepfs.ufs@gmail.com)

Version: 2 Date: 22 Jun 2017

Author’s response to reviews:

Dear Dr. Domenico Capone,

Thanks for your assistance in our manuscript entitled: “Evaluation of pharmacotherapy complexity in residents of long-term care facilities: a cross-sectional descriptive study” (PHAT-D-17-00050). We are aware of the minor revisions, and we carefully considered all corrections.

Kind regards,

Dr. Divaldo Pereira Lyra Junior, PhD.

Laboratory of Teaching and Research in Social Pharmacy (LEPFS), Federal University of Sergipe, São Cristóvão, SE, Brazil.

Address: Cidade Universitária “Prof. José Aloísio Campos”, Jardim Rosa Elze, São Cristóvão, CEP: 49100-000, Brazil.

E-mail: lyra_jr@hotmail.com; lepfs.ufs@gmail.com

Phone/Fax: +5579991925577